 Vc/6-May-04
Week 19 starting 03 May 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for now & next few days : 

	Bench
	Name
	Monday 03/05  Afternoon status
	 

	TBB1
	3083
	New  Arrival
	 with Anti C

	TBB2
	_
	New ?
	 

	TBC1
	MBBL3089
	PT 10 Min. NRJ 11850 A
	 

	TBC2
	SSS015
	EX1 Extra Test
	 

	TBD1
	MBBL2049
	ICS 3 Traveller after warm test
	 

	TBD2
	_
	New ?
	 

	TBE1
	MBBL3123
	PT 5.2 1.5 KA Quench HF
	 

	TBE2
	MBBL1110
	PT 12 Warm Up
	 

	TBF1
	MBBL1043
	ICS 3 Traveller after warm test
	 with Anti C

	TBF2
	_
	New ?
	 


Test Coordinator this week: J Miles ; MTM support on call  Guy:  (16-3602).

4 Dipoles  Disconnected after tests last week    ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· B2 mech. work for legs completed on Monday ? ( J Halik)
· Preparation of D Cluster for SSS  ( J.Halik?)
· General Problems last week :
 SSS work : 

Nair + Sridhar , SSW etc etc..  
	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 27_N
	TBC1
	During IAP permanent external trigger on potaim card.


	ICS
	
	Ewald informed.  One taps not connected. ICS informed. The team opened the magnet and found that the external lead connection was not OK. Solved 

	Tuesday 27_A
	ALL
	No water in SM18 
	
	
	TCR called. OK after 1 hour.

	Tuesday 27_A
	TBC2
	Communication between PLC and power converter not working on SSS.  
	AB/CO?

AB/PO?


	
	Ewald informed. Recalibration of the power supply asked by E.Garde and authorised by Nicolay?

	Tuesday 27_A
	TBC2
	Discrepancy between T read in TEMA and in through cryogeny.
	MAS
	
	Thermometer calibration of the SSS not present in MTF?

	Tuesday 27_A
	TBF2, TBE2
	Magnets 3104 and 3110: TT821 sensor coefficient not present in MTF.
	MAS
	
	Ewald informed Stéphane. He contacted directly P.Lienard. Problem solved for the future.

	Wednesday 28_M
	TBC2
	SSS powered. Ramp to 5 kA but SPA.

PLC reading reports “cryo loss”.
	ACR
	
	Pressure on the lines going down due to restart of cold compressor tests in BA7?

	Thursday 29_A
	TBB2
	QH control crate not scan because sticker missing. 
	MTM
	
	Equipment entered manually by Ewald.

	Friday 30_A
	TBD1
	3072: Quench at 4.2 K to warm up. Faulty quench.

Flat signals in HF off: D1 L-U D1-D2…

 
	MTM
	HF
	Stéphane checked the quench. Apparently consistency in the direct signals picture. He advised Ewald to check the HF recording 

	Friday 30_A
	TBD2
	SPA occurred at 8629 A. 
	MTM?

ACR? ABPO?
	
	

	Friday 30_A

Sunday 2_M
	TBB1
	Switch from to CC position possible. PLC error during IAP

Rotary switch command box is not working.

(no light, no answer when you press the buttons)


	MTM


	
	Ewald called.

He managed to solve punctually the problem.

Guy will inform Giorgio Monday for an action.

	Saturday 1_N
	TBF1
	IAP at cold: HF and LF armed. Only LF received the triggers. HF acquisition problem


	AB/CO
	HF
	Reset the crates and restart TEMA. Not OK

Ewald called. He changed one card and reset the MXI.

	Saturday_1_M

Sunday 2_M
	
	Video TV display not working.


	
	
	Clear instructions are needed.

Call Dirk Bakker?

	Sunday 2_N
	TBB1
	STMS not updated because SM18 Database is not working


	?


	
	








